Author’s response to reviews

Title: Paediatric UK Demyelinating Disease Longitudinal Study (PUDDLS)

Authors:

Michael Absoud (michael.absoud@childdemyelination.org.uk)
Carole Cummins (c.l.cummins.20@bham.ac.uk)
Wui K Chong (chongk@gosh.nhs.uk)
Christian Degoede (christian.degoede@lthtr.nhs.uk)
Katharine Foster (Katharine.foster@bch.nhs.uk)
Roxana Gunny (gunnyr@gosh.nhs.uk)
Cheryl Hemingway (heminc@gosh.nhs.uk)
Philip Jardine (philip.jardine@bristol.ac.uk)
Rachel Kneen (Rachel.kneen@alderhey.nhs.uk)
Marcus Likeman (Marcus.likeman@nbt.nhs.uk)
Ming J Lim (ming.lim@gstt.nhs.uk)
Mike Pike (mike.pike@orh.nhs.uk)
Naomi Sibtain (Naomi.sibtain@nhs.net)
William P Whitehouse (William.whitehouse@nottingham.ac.uk)
Evangeline Wassmer (Evangeline.wassmer@bch.nhs.uk)

Version: 3 Date: 14 July 2011

Author’s response to reviews:

Dear Ms Rachel Neilan,

MS: 1826082349551067

Paediatric UK Demyelinating Disease Longitudinal Study (PUDDLS).

Thank you for your letter email dated the 7th July 2011 requesting more information, specifically to:

1) provide a document which shows the total amount of funding from both the Multiple Sclerosis Society and Action Medical Research.

2) revise the Authors' contributions section so that it contains the specific contributions of each author.

3) to revise the Abstract to give more general information about the context of the study and the contributions it might bring.

We have attached a document which shows the total amount of funding from both the Multiple Sclerosis Society and Action Medical Research. We have revised the Authors' contributions section so that it contains the specific contributions of each author. We have also revised the Abstract to give more
general information about the context of the study and the contributions it might bring.

We look forward to hearing from you.

Yours Sincerely,

PUDDLS authors